Author’s response to reviews

Title: The antifibrotic drug pirfenidone inhibits spondyloarthritis fibroblast-like synoviocytes and osteoblasts in vitro

Authors:

Julie Stougaard (jkl@biomed.au.dk)
Søren Lomholt (soren.lomholt@gmail.com)
Pernille Ommen (pernille.ommen.andersen@post.au.dk)
Jens Kelsen (jenskels@rm.dk)
Tue Kragstrup (kragstrup@biomed.au.dk)

Version: 5 Date: 05 Sep 2018

Author’s response to reviews:

Response letter #5

Thank you for the fast response.

Editor Comments:

The manuscript has been much improved. I have one last comment to you.

I wrote the same comment to you last time, but the order of figures should be placed in the order you mention in the manuscript. For example, figure 2D should be changed as figure 2B, since it is the second figure you mention in the paragraph. Also, please change the figure legend accordingly.

Answer. Thank you. This has been corrected.